Author’s response to reviews

Title: Nomogram for individually predicting overall survival in rectal neuroendocrine tumours

Authors:

Xingyu Feng (494368185@qq.com)
Gengzhou Wei (gzwei068273@163.com)
Wei Wang (wangwei@sysucc.org.cn)
Yu Zhang (13430357659@163.com)
Yujie Zeng (sumsyorky@hotmail.com)
Minhu Chen (chenminhu@vip.163.com)
Ye Chen (yechen@smu.edu.cn)
Jie Chen (chen0jie@hotmail.com)
Zhiwei Zhou (zhouzhw@sysucc.org.cn)
Yong Li (sywcwktg@126.com)

Version: 3 Date: 28 Jul 2020

Author’s response to reviews:

Dear editor
This is a clean version of my manuscript and do not include track changes or highlighting. And I add the Authors’ Information in the Declarations section at line 6, page 13 according to the submission guidelines.

Editor
Reading the last version of the manuscript I noted some typos and english grammar issues. Please have the manuscript verified by a fluent/native speaking person to enhance the quality of written english.

Response: Thanks for your suggestion. The English written of the manuscript had been edited by American Journal Experts. I will uploading the certificate as the Supplementary Material.

Reviewer 1
Thank you very much for providing a revised version of the manuscript.
Response: Thank you for your approval of our article.

Reviewer 2:
1. Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs,
supporting materials or other aspects of your report which cannot be included in a text format. Please overwrite this text when adding your comments to the authors.
Response: Thanks for your suggestion. I have included all comments for the authors in the box and do not submit any attachment except for the manuscripts, graphs, figures and supporting materials.
2. With the apparent issues with statistical analysis addressed I have no question.
Response: Thank you for your approval of our article.